Author’s response to reviews

Title: Identifying older diabetic patients at risk of poor glycemic control

Authors:

Raffaele Antonelli Incalzi (raffaele_antonelli@rm.unicatt.it)
Andrea Corsonello (andrea_corsonello@dada.it)
Claudio Pedone (claudio_pedone@rm.unicatt.it)
Francesco Corica (coricaf@www.unime.it)
Luciana Carosella (luciana_carosella@rm.unicatt.it)
Bruno Mazzei (mazzaib@katamil.com)
Francesco Perticone (perticone@unicz.it)
Pierugo Carbonin (pierugo_carbonin@rm.unicatt.it)

Version: 4 Date: 21 Aug 2002

To the Editor:
1. The corresponding author has been included on title page
2. Abstract has been shortened to 250 words
3. References are now cited in the text using consecutive numbers in square brackets
4. References list has been provided in BMC style
5. Figure 1 has been cropped

To the referee:
The final sentence of the first paragraph of the Discussion has been changed as suggested by eliminating the expression "versus the latter outcome".